%%%%%%%%%%%%%% Presentation Template %%%%%%%%%%%%%%%%%%%%
\documentclass[a4,12pt]{article}
\usepackage{epsfig}

\setlength{\topmargin}{3mm}
\setlength{\textheight}{229mm}
\setlength{\evensidemargin}{-3mm}
\setlength{\oddsidemargin}{-3mm}
\setlength{\textwidth}{154mm}

\thispagestyle{empty}

\begin{document}

%\input{helge_header}
\input{hopfentation_header}
{\scriptsize \bf 04/06}
\vspace{4ex}

\centerline{\Huge\bf Our solutions}
\vspace{1cm}

\Large

\begin{itemize}
 \item {\bf Object separation}
 \begin{itemize}

	\item{Finding Edges:}
	\begin{figure}[!ht]
	\begin{center}
	\includegraphics[width=7cm]{sobel3.eps}
	\end{center}
	\end{figure}

	\item{Thinning the edges; Following the edges and separating the objects:}
	\begin{figure}[!ht]
	\begin{center}
	\includegraphics[width=7cm]{comb3.eps}
	\end{center}
	\end{figure}  
 \end{itemize}

 \item Transformation to activation pattern
 \begin{itemize}
  \item Conversion of picture to 1-dimensional waveform
	\begin{figure}[!ht]
	\begin{center}
	\includegraphics[width=8cm]{tomo.eps}
	\end{center}
	\end{figure}  
 \end{itemize}
\end{itemize}
\clearpage

\end{document}
